Author's response to reviews

Title: Academic doctors' views of complementary and alternative medicine (CAM) and its role within the NHS: an exploratory qualitative study

Authors:

Nita Maha (nitamaha@nhs.net)
Alison Shaw (ali.heawood@bristol.ac.uk)

Version: 3 Date: 22 May 2007

Author's response to reviews:

Dear Editor

Here are the following changes to my manuscript:

1. The title has been changed to be the same in the manuscript and submitted version.
2. The introduction title has been renamed background.
3. The website addresses have been cited in the recommended format.
4. The titles have been removed from figure 1 and 2 as requested.

Many thanks

Dr N Maha